Author's response to reviews

Title: Cognitive, fear-reducing information or individual symptom-based physical training in chronic LBP. A pragmatic, randomized, controlled trial with 1-year follow-up.

Authors:

Pia H Sorensen (piahavn@hotmail.dk)
Tom Bendix (tbendix@health.sdu.dk)
Claus Manniche (claus.manniche@slb.regionsyddanmark.dk)
Lars Korsholm (korsholmlars@gmail.com)
Dorte Lemvigh (dorte.lemvigh@slb.regionsyddanmark.dk)
Aage Indahl (aage.indahl@rehabilitering.net)

Version: 9 Date: 2 September 2010

Author's response to reviews: see over
Dear BMC

I have now edited the things you wants - I think. Comments follows the various points:

Formatting changes to be made:
---------------------------------

Major revisions (we require the author to make these changes)

Abstract: please ensure that the Abstract is present in manuscript and matches exactly what is in the submission system

Her I have the same problems as last time with the title: It has been changed during the review process. So ... what do you mean with “the submission system” ??

Table 5: we notice that table 5 is not cited in the main text. Please either cite table 5 or remove table 5 from the submission.

There was a printing matter Now Table 5 is in the text. Sorry!

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Email address: please include email addresses for all the authors on the title page, using the following format:
AB: abcd@institution.ac.uk
EF: efgh@generic.co.uk
IJ: IJKL@corporation.com

I think that the new style is what you want ... and what we have seen in various articles from your journal.

Affiliations: For affiliations that have been up-dated since writing the paper, please include the following text "Current address:" Please avoid the use of "Present address:“, "New address:"

done

Table: please remove any table placement text.

I don’t understand this: We do not have Tables in the text. We have indicated where they optimally should be placed. ... Thus ...??

Authors' contributions: please ensure this is formatted into a paragraph rather than a list or in bullet points.

done

References: the reference list should contain up to 30 authors' names for each citation. The term 'et al.' should not be used if there are fewer than 30 authors.

Have just heard from Robyn Hughes that 6 authors + et al is OK although.
Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

These your two last paragraph heve been considered.

Please make sure you use the latest version of the main file as we have made several formatting changes and using a previous unformatted version will delay publication.

Also I have sent the payment not to our “minister of finance” 😊

Best regards
Tom